Author's response to reviews

Title: San Huang Shel Shin Tang beta-cyclodextrin complex augmented the hepatoprotective effects against carbon tetrachloride-induced acute hepatotoxicity in rats

Authors:

Wei-Syun Hu (duodenum222000@yahoo.com.tw)
Wei-Jen Ting (tim_sigeen@hotmail.com)
Dennis Jine-Yuan Hsieh (djh@csmu.edu.tw)
Hsi-Hsien Hsu (u596@mmc.edu.tw)
Yueh-Min Lin (YuehMinLin@outlook.com)
Chia-Yao Shen (x00003061@meiho.edu.tw)
Chung-Ho Chang (changch@nhri.org.tw)
Viswanadha Vijaya Padma (padma.vijaya@gmail.com)
Yuhsin Tsai (yhtsai@mail.cmu.edu.tw)
Chih-Yang Huang (cyhuang@mail.cmu.edu.tw)

Version: 6
Date: 1 December 2015

Author's response to reviews: see over
Reviewer's report
Title: San Huang Shel Shin Tang beta-cyclodextrin complex augmented the hepatoprotective effects against carbon tetrachloride-induced acute hepatotoxicity in rats
Version: 4 Date: 8 October 2015
Reviewer: Shanthi Palanivelu

Reviewer's report:
Although the authors have made most of the revisions suggested, the revision of fig 1 is yet to be carried out. However, I am still doubtful about the accuracy of histopathological and special stain interpretation. This article may be sent to another expert for evaluation.

Responds: We reexamined the H&E and MT staining slices. The resolution and results were more clear in the revised figure 1.

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Needs some language corrections before being published
Responds: Some language errors were corrected in revised manuscript.

Statistical review: Yes, but I do not feel adequately qualified to assess the statistics.

Declaration of competing interests:
'I declare that I have no competing interests